Author's response to reviews

Title: A cluster randomized-controlled trial of a community mobilization intervention to change gender norms and reduce HIV risk in rural South Africa: Design, implementation and baseline findings

Authors:

Audrey Pettifor (apettif@email.unc.edu)
Sheri A. Lippman (Sheri.Lippman@ucsf.edu)
Amanda Selin (aselin@UNC.EDU)
Dean Peacock (dean@genderjustice.org.za)
Ann Gottert (Annagottert@live.unc.edu)
Suzanne Maman (Suzannmaman@email.unc.edu)
Dumisani Rebombo (dumisani@genderjustice.org.za)
Chirayath M. Suchindran (suchi@bios.unc.edu)
Rhian Twine (Rhian.Twine@wits.ac.za)
Kathryn Lancaster (Lancaklanc@live.unc.edu)
Tamu Daniel (tamudaniel@gmail.com)
F.Xavier Gómez-Olivé (F.Gomez-OliveCasas@wits.ac.za)
Kathleen Kahn (Kathleen.Kahn@wits.ac.za)
Catherine MacPhail (cmacphai@une.edu.au)

Version: 2
Date: 26 June 2015

Author's response to reviews:

Per the comments we have removed the baseline results of the paper and only describe the study and intervention design.